Author's response to reviews

Title: Conservation of Resources Theory and Research Use in Health Systems

Authors:

Celeste Alvaro (Celeste.Alvaro@dal.ca)
Renee F Lyons (RLyons@bridgepointhealth.ca)
Grace Warner (Grace.Warner@dal.ca)
Stevan E Hobfoll (Stevan.Hobfoll@rush.edu)
Patricia J Martens (Pat_Martens@cpe.umanitoba.ca)
Ronald Labonte (rlabonte@uottawa.ca)
E. Richard Brown (erbrown@ucla.edu)

Version: 2 Date: 2 December 2009

Author's response to reviews:

As per the editorial team's request, this version of the manuscript has been significantly reduced in length by nearly 2200 words (from 8085 words to 5887 words). The editorial team provided valuable suggestions on ways to do so.